Author's response to reviews

Title: Coordination and integration in primary community care networks (PCCNs): a national demonstration project in Taiwan

Authors:

Blossom YJ Lin (yenju1115@hotmail.com)

Version: 2 Date: 22 December 2006

Author's response to reviews: see over
Dear Editor-in-chief:

Taiwan primary community-based care network (PCCN) demonstration project was a national project which have launched by the Bureau of National Health Insurance since 2003 and still processing now. This article I am going to submit is the first one of the series of my articles, looking at the Taiwan PCCN demonstration project from the organizational research perspectives. This article was tried to develop some dimensions to understand how the network members have done for integration/coordination efforts. In addition, I am working on the following articles which examined the effectiveness. I thought all these studies have paid more explicit attention to the organizational characteristics and processes of coalitions. I hope your journal could consider my studies as the readers’ interests.

Sincerely,

Blossom Yen-Ju Lin